Author's response to reviews

Title: A comparison of survival outcomes and side effects of toremifene or tamoxifen therapy in premenopausal estrogen and progesterone receptor positive breast cancer patients: a retrospective cohort study

Authors:

Fengxi Su Prof (fengxisu@vip.163.com)
Ran Gu Dr (ccaibai@qq.com)
Weijuan Jia M.D. (jaweijuan@yahoo.com.cn)
Yunjie Zeng Dr (zyj81332590@126.com)
Nanyan Rao Prof (raonany@126.com)
Yue Hu Dr (fionahu126@126.com)
Shunrong Li Dr (lison2000@163.com)
Jiannan Wu Dr (king8702@163.com)
Liang Jin Dr (jini2001@qq.com)
Lijuan Chen (337330348@qq.com)
Erwei Song Prof (songew@mail.sysu.edu.cn)
Meijun Long Dr (meijunlong@yahoo.com.cn)
Kai Chen Dr (ckaichen@126.com)
Qiaozhen Xiao (ccaibai@sina.com)
Mei Wu (wumeixiaopig@163.com)
Lili Chen (ccaibai@163.com)

Version: 13 Date: 12 February 2012

Author's response to reviews:

Dear Dr. Chap,

We would like to thank you for your constructive comments regarding our manuscript. Based on your suggestions, we wish to re-submit our manuscript entitled, ‘A comparison of survival outcomes and side effects of toremifene or tamoxifen therapy in premenopausal estrogen and progesterone receptor positive breast cancer patients: a retrospective cohort study’.

Among the changes made to the manuscript, the heterogeneity between the two groups has been discussed in further depth, and HR and 95% confidence interval data has been included to Tables 4 and 5.

We have also enlisted the services of a professional editor to ensure that the
manuscript text is of the quality necessary for publication.

We thank you for considering our paper for publication and look forward to your reply.

Yours sincerely,

Drs Fengxi Su & Erwei Song